vc/21-Oct-03
Week 43 starting 20 Oct 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

Today(21/10/03)
	Forward look,   & Next Magnet 

	TBA1
	3042
	Leak test finished –remove 3042
	Bench de-activated – preparation for D bench

“Park” here 2001 ( moved from D bench )

	TBA2
	1020
	Disconnect & Remove
	 Next Magnet 1049

	TBC1
	3044
	Disconnect & Remove
	Next Magnet 2024 

	TBC2
	SSS1
	cold
	Diode Cold tests then warm up & remove; bring new SSS

	TBE1
	2028
	Connect & cooldown
	START COLD TESTS

	TBE2
	1050
	Cold 
	Need to undergo Thermal Cycle

	TBF1
	2002 
	cold
	Commissioning/powering tests

	TBF2
	2025
	Warm 
	Next Magnet 3058 : 

as first time Usage of F bench for Operation




Test Coordinator this week:  Olav Berrig;  MTM  Tech. on call:  (16-3602) : Guy Deferne
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Loading up time &  AQA Analyses files(Daniel)

2. Traveller Followup – “I” shift 0830-1730

3. F Bench Commissioning 

4. Interactions in Traveller & Info  ( Q-Table, HV Tests)

5. AOB

Current Bench Status:

SM18 OPERATION WEB PAGE 
	Bench
	Magnet
	Test
	Temp.
	Time
	Current

	TBA1
	_
	_
	 
	10/10/2003 9:20:27 AM
	0


	TBA2
	MBBR1020
	Finished Ready to go
	 
	10/20/2003 6:31:23 PM
	0


	TBB1
	_
	_
	 
	 
	0


	TBB2
	_
	_
	 
	 
	0


	TBC1
	MBAL3044
	Finished Ready to go
	 
	10/20/2003 6:31:40 PM
	0


	TBC2
	SSS001
	_
	 
	10/19/2003 12:35:33 PM
	0


	TBD1
	_
	_
	 
	 
	0


	TBD2
	_
	_
	 
	 
	0


	TBE1
	MBAR2028
	Prep6.3 QH Measure
	300 K
	10/21/2003 10:15:00 AM
	0


	TBE2
	MBBL1050
	PT30 HV Thermal Cycle
	Warming
	10/21/2003 11:27:52 AM
	0


	TBF1
	MBBR2002
	PT3 HV Cold Test
	1.9 K
	10/20/2003 1:57:42 AM
	0


	TBF2
	MBBR2025
	Finished Ready to go
	300 K
	10/21/2003 10:00:00 AM
	0


· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 13_A
	TBF2 Training quench ; Quench at 10437A.

Je ne sais pas ou est passe le fichier de ce quench : L0310142027.a02
	
	
	

	Tuesday 13_N
	TBA2: Final HV test at cold.

There is no test sequence for HV at cold with Power leads connected to power converter.

Test launched and sequence 4 onwards selected after run recovery.
	
	
	

	Wednesday 14_M
	TBE1: SSL. Server lost on sunmta26 
	
	
	restarted OK

	Thursday 15_M
	TBE2: IAP @ Warm ; HF Discharge; The test could not launched as magair300_x2 and mag300_x2  interlock signals are not clearing. Equipment support was contacted and Richard is on the job.


	
	
	- Cryo work on Cluster D took the compressed air supply from Cluster E, causing the interlock. OK after cryo had finished.

	Thursday 16_N
	TBF2: No HV test at cold and no Thermal cycle because the rotary switch is blocked
	
	
	

	Friday 17_A
	TBA2 - DISCONNECTION start for ICS

TBE1 - CONNECTION of 2028  CONTINUE for ICS

 FOR BOTH POINTS ICS WILL NOT COME during weekend & so now WE ARE WAITING FOR Them for 2 days !!
	
	
	ICS will soon (November) have 3 shifts i.e. 7/7 & 24/24

	Friday 17_A
	*** Cold Tests on TBE2 (  1.9 K  tomorrow (saturday)  : Here Shafts need to be installed but there is a problem due to anticryostat being too warm for reasons unknown. Patrick Legrand informed
	
	
	

	Friday 17_A
	TBC1 : Two Q Loc cards not getting compensated.

D1_L-U potaim card was not getting compensated. Exchanged the card from A1 crate as suggested by Richard. Problem solved. The card is kept in the almirah in the control room on which rota is pasted.
	
	
	

	Saturday 18_A
	TBE2: IAP at cold. Vgas_Int_lead exceeds from 10 mv( offset) to 10.7 mv but could not be adjusted. Some errors in Qloc cards could not be adjusted.
	
	
	

	Saturday 18_N
	TBE1:MM position problem CPU encoder error.

As per the instruction from Richard, we reset the CPU's by switching off the power supply. 

The MM position measurement could be carried out without any further problem.
	
	
	In case of MM problems, do not switch off the MMP program, but go to the hardware panel. In the hardware panel, it is possible to switch off/on the VME crates.

It is not necessary to switch off the MMP program.

	Saturday 18_N
	TBC1: We wanted to carry out the HV insulation test at cold ; However, "Run Sequence " did not succeed for the first test. It was not even giving the voltage of 650 V. The megger was giving red beeps and displaying "burn". We checked all cables and found to be alright. We found that the ground cable from the magnet to the ground bus cable had not been properly connected. We have connected it by twisting the cables but this requires improvement. Though this may not be the reason for not able to proceed further, this requires to be improved. 

After this, the megger was able to give voltage but once the measurement for that step was over, the message "Capacitance values exceed exspectation " came which could not be cleared despite repeated attempts . Hence we could not proceed further. 
	
	
	We tried to do tests step-by-step; Steps involving dipoles did not succeed while others could be done.

	Saturday 18_N
	TBE2: Minimum Energy Quench at 3000 A; 1.9 K. Vquench is between 700 and 710 volts.

The data was analyzed for the diode test as required by Eric Flock. A print-out of the same is kept in the log. However, we did not understand how to calculate the diode joint resitance. Next shift may talk to Eric Flock Or Nick for further instructions on this. THE DESCRIPTION OF THIS TEST IS KEPT IN "Quench Procedures" FOLDER(Name : "Brief Description of the method used to measure the diode contact resistances")
	
	
	

	Sunday 19_M
	TBC1: Mimimun Energy @6.5kA +RRR, Vquench=540V

NOTE: LF configuration file is not found.
	
	
	Remember to wait for the LF data to be transferred, before doing the RRR test.

	Sunday 19_M
	TBE1: We suspect some problem with anticryostat A1L showing a 0 value since 7:00 pm

Problem is confirmed. Anticryostat temperature is 58 C.

Richard, Nick informed

P. Legrand contacted by Nick. Didier Cotte contacted by myself, he will come.

Everybody is here. Didier removes the shafts. P. Legrand found a bad contact on A2L connector.
	
	
	

	Monday 20_M
	TBF2: IAP @ warm did not work from TEMA. Giorgio believes it is a problem with the PLCSE. He has sent an e-mail to Claude Dehavay to solve it.
	
	
	

	Monday 20_A
	TBE2:  Terminal screen kaput - switches off by itself. Magnus Bjork called by Adriaan.
	
	
	


	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 14_M


	TBA2:  MM LHC export blocked
	MTM
	MMP
	Problem under investigation.

	Tuesday 14_M


	TBF2:  Quench at 6.5 kA

MM positioning instead of Q.loc
	MTM
	
	

	Tuesday 14_M


	TBF2: Training quench 2

File L0310142027.a02 lost
	MTM
	
	

	Tuesday 14_M


	TBA2: No setting HV test final at cold with current leads connected.
	MTM
	
	Test launched and start and sequence 4 selected after run recovery (instruction from Richard).

	Tuesday 14_M


	TBA2: No setting HV test final at cold with current leads connected.
	MTM
	
	Test launched and sequence 4 selected after run recovery (instruction from Richard).

	Wednesday 15_N


	TBF2: No data for anti-cryostat in the control room.
	ACR?
	
	

	Wednesday 15_N


	TBF2: Cryo problem
	ACR
	
	

	Thursday 16_M


	TBE2: (IAP @ warm)  HF discharge could not be launched. Mag_air_300_X2, Mag_300_X2 interlocks.
	ACR
	
	Compress air supply from cluster E used for work in cluster D?

	Friday 17_M


	TBF2: Anti-cryostat at –180  C due to a lost of vacuum inside the cold mass!

(information from P.Legrand)
	ACR
	
	


	
	
	
	
	


